\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at195d.txt}{\rawdatadir/at195d.txt}~Aminallah (1993)~\cite{m_amin93}
\W\label{at195d}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at195d~\protect\cite[Aminallah (1993)]{m_amin93}\label{at195d}}\\
\T\input{\datadir/at195d.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & CH4 \\
Sub-Category: & cylindrical, exploding wire & Oxidizer: & O2 \\
Initial Pressure: & 100 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & .7-1.3 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at195e.txt}{\rawdatadir/at195e.txt}~Aminallah (1993)~\cite{m_amin93}
\W\label{at195e}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at195e~\protect\cite[Aminallah (1993)]{m_amin93}\label{at195e}}\\
\T\input{\datadir/at195e.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & CH4 \\
Sub-Category: & cylindrical, exploding wire & Oxidizer: & O2 \\
Initial Pressure: & 120 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & .7-1.3 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at177a.txt}{\rawdatadir/at177a.txt}~Beeson (1991)~\cite{h_bees91}
\W\label{at177a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at177a~\protect\cite[Beeson (1991)]{h_bees91}\label{at177a}}\\
\T\input{\datadir/at177a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & CH4 \\
Sub-Category: & spherical & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at210a.txt}{\rawdatadir/at210a.txt}~Bull (1976)~\cite{d_bull76}
\W\label{at210a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cccc}
\T\caption{at210a~\protect\cite[Bull (1976)]{d_bull76}\label{at210a}}\\
\T\input{\datadir/at210a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & CH4 \\
Sub-Category: & spherical, high explosive & Oxidizer: & O2 \\
Initial Pressure: & 101.3 kPa & Diluent: & N2 \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at209a.txt}{\rawdatadir/at209a.txt}~Desbordes (1973)~\cite{d_desb73}
\W\label{at209a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at209a~\protect\cite[Desbordes (1973)]{d_desb73}\label{at209a}}\\
\T\input{\datadir/at209a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & CH4 \\
Sub-Category: & cylindrical & Oxidizer: & O2 \\
Initial Pressure: & 100 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at210b.txt}{\rawdatadir/at210b.txt}~Kogarko (1965)~\cite{s_koga65}
\W\label{at210b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cccc}
\T\caption{at210b~\protect\cite[Kogarko (1965)]{s_koga65}\label{at210b}}\\
\T\input{\datadir/at210b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & CH4 \\
Sub-Category: & spherical, high explosive & Oxidizer: & O2 \\
Initial Pressure: & 101.3 kPa & Diluent: & N2 \\
Initial Temperature: &  K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at79a.txt}{\rawdatadir/at79a.txt}~Matsui (1979)~\cite{h_mats79}
\W\label{at79a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at79a~\protect\cite[Matsui (1979)]{h_mats79}\label{at79a}}\\
\T\input{\datadir/at79a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & CH4 \\
Sub-Category: & spherical & Oxidizer: & O2 \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & .7-1.7 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at195c.txt}{\rawdatadir/at195c.txt}~Nicholls (1979)~\cite{j_nich79}
\W\label{at195c}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at195c~\protect\cite[Nicholls (1979)]{j_nich79}\label{at195c}}\\
\T\input{\datadir/at195c.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & CH4 \\
Sub-Category: & cylindrical, high explosive & Oxidizer: & O2 \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 298 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at115.txt}{\rawdatadir/at115.txt}~Wolanski (1981)~\cite{p_wola81}
\W\label{at115}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at115~\protect\cite[Wolanski (1981)]{p_wola81}\label{at115}}\\
\T\input{\datadir/at115.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & CH4 \\
Sub-Category: & planar & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & .8-1.4 \\
\end{tabular}
\end{center}

